

## OPEN ACCESS

**Citation:** Seoighe C, Scally A (2017) Inference of Candidate Germline Mutator Loci in Humans from Genome-Wide Haplotype Data. PLoS Genet 13(1): e1006549. doi:10.1371/journal.pgen.1006549

**Editor:** Guy Sella, Columbia University Department of Systems Biology, UNITED STATES

Received: July 18, 2016

Accepted: December 20, 2016

Published: January 17, 2017

**Copyright:** © 2017 Seoighe, Scally. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data from phase 3 of the 1000 genomes project are available from ftp:// ftp.1000genomes.ebi.ac.uk/vol1/ftp/release/ 20130502/. Data from the Genome of the Netherlands consortium are available via the Data Access Committee of the Genome of the Netherlands (see http://www.nlgenome.nl/).

**Funding:** The authors received no specific funding for this work.

**Competing Interests:** The authors have declared that no competing interests exist.

**RESEARCH ARTICLE** 

# Inference of Candidate Germline Mutator Loci in Humans from Genome-Wide Haplotype Data

#### Cathal Seoighe<sup>1,2</sup>\*, Aylwyn Scally<sup>3</sup>

1 School of Mathematics, Statistics and Applied Mathematics, NUI Galway, Galway, Ireland, 2 Institute of Infectious Disease and Molecular Medicine, University of Cape Town, Cape Town, South Africa, 3 Department of Genetics, University of Cambridge, Cambridge, United Kingdom

\* Cathal.Seoighe@nuigalway.ie

## Abstract

The rate of germline mutation varies widely between species but little is known about the extent of variation in the germline mutation rate between individuals of the same species. Here we demonstrate that an allele that increases the rate of germline mutation can result in a distinctive signature in the genomic region linked to the affected locus, characterized by a number of haplotypes with a locally high proportion of derived alleles, against a background of haplotypes carrying a typical proportion of derived alleles. We searched for this signature in human haplotype data from phase 3 of the 1000 Genomes Project and report a number of candidate mutator loci, several of which are located close to or within genes involved in DNA repair or the DNA damage response. To investigate whether mutator alleles remained active at any of these loci, we used *de novo* mutation counts from human parent-offspring trios in the 1000 Genomes and Genome of the Netherlands cohorts, looking for an elevated number of *de novo* mutations in the offspring of parents carrying a candidate mutator haplotype at each of these loci. We found some support for two of the candidate loci, including one locus just upstream of the *BRSK2* gene, which is expressed in the testis and has been reported to be involved in the response to DNA damage.

### Author Summary

Each time a genome is replicated there is the possibility of error resulting in the incorporation of an incorrect base or bases in the genome sequence. When these errors occur in cells that lead to the production of gametes they can be incorporated into the germline. Such germline mutations are the basis of evolutionary change; however, to date there has been little attempt to quantify the extent of genetic variation in human populations in the rate at which they occur. This is particularly important because new spontaneous mutations are thought to make an important contribution to many human diseases. Here we present a new way to identify genetic loci that may be associated with an elevated rate of germline mutation and report the application of this method to data from a large number

of human genomes, generated by the 1000 Genomes Project. Several of the candidate loci we report are in or near genes involved in DNA repair.

#### Introduction

The rate of germline mutation is a key parameter in molecular evolution and population genetics. As the ultimate source of genetic novelty, germline mutations provide the raw material on which selection acts and the basis for genetic drift over time. Mutation rates are known to differ substantially between species [1], and in eukaryotes, the single nucleotide mutation rate, fundamental to many demographic and evolutionary analyses, ranges over two orders of magnitude [2]. Methods to estimate the rate of *de novo* mutation predate the knowledge that DNA carries hereditary information. If the frequency of a deleterious allele is in mutation-selection balance, the rate at which deleterious alleles are removed through selection is equal to the rate at which novel alleles arise through mutation. This idea was used by Haldane to provide an indirect estimate of the rate of spontaneous haemophilia from prevalence estimates [3]. Subsequent estimates incorporated knowledge of the physical size of the locus (or more precisely, the number of target nucleotides giving rise to the phenotype of interest) to obtain per-base and per-generation mutation rate estimates (e.g. [4, 5]).

More recently, whole genome resequencing methods have been used to obtain direct measurements of the human mutation rate from parent-offspring trios [6–9]. As well as enabling measurement of the genome-wide mutation rate, these studies have opened up the possibility of investigating factors contributing to variation in mutation rate between individuals. A study of 78 Icelandic parent-offspring trios estimated that 97% of variation in the number of *de novo* mutations called in offspring could be explained by the age of the father [8], and other wholegenome studies have shown similar, albeit weaker, relationships between parental age and germline mutation rate [10–12].

Genetic and environmental factors may also influence the rate of germline mutation [10, 13], and a number of recent studies point to differences in the germline mutation spectrum between human populations [13–15], consistent with a genetic contribution to variation in germline mutation. Indeed, several cancer-associated germline human mutations are known to affect genes involved in DNA proofreading and mismatch repair, increasing cancer risk through an elevated rate of somatic mutation [16, 17]. For example, Lynch syndrome, which is associated with a very high lifetime risk of development of cancer of the colon and several other organs, results from germline mutations in a number of mismatch repair genes [18]. Although the effects of such mutations on the rate of germline mutation in human have not yet been determined, deficiency in mismatch repair is known to result in an elevated per generation mutation rate in yeast [10].

Here we investigate the feasibility of detecting genetic polymorphism associated with increased germline mutation rate by looking for an increase in the number of derived alleles in haplotypes carrying the mutator allele. Using simulation, we show that a mutator allele can result in a localized peak in the numbers of derived alleles in a subset of haplotypes, against a background of haplotypes with typical numbers of derived alleles. This is because in the region linked to the mutator allele, haplotypes containing it are always subject to the elevated mutation rate, whereas other haplotypes are only affected when they occur together with the mutator allele in a heterozygote, and this will occur only rarely if the mutator allele is rare. Detecting this pattern depends on persistence of the mutator allele for a large number of generations, and we discuss the likelihood of this, given estimates of the selective disadvantage of a

mutator allele. We search for this pattern in data from the 1000 Genomes Project (G1K) and report and characterise a number of candidate mutator loci. For a subset of the candidate loci the highly derived haplotypes, characteristic of a mutator allele, were found among parents of two human trio datasets obtained from the Genome of the Netherlands (GoNL) [19] and G1K [20] projects and in each case we tested for an elevated number of *de novo* mutations in the off-spring of parents carrying a putative mutator haplotype.

#### **Results and Discussion**

We hypothesized that mutations that increase the rate of germline mutation in human populations may be detectable from genome-wide polymorphism data. Consider a population with a genetic variant at a locus that increases the germline mutation rate by a factor  $\phi$  (in heterozygotes). If the variant arose *g* generations before the present, we can estimate the number of additional derived alleles, per site, that are expected in haplotypes containing the mutator allele over haplotypes with the wild-type allele as  $g\mu(\phi - 1)$ , where  $\mu$  is the wild-type mutation rate. Let  $\rho$  be the rate of recombination per generation, which, for simplicity, we assume is uniform across the genome. The lengths of the genomic segments to the right and left of the mutator locus that have been free from recombination for *g* generations are then exponentially distributed with mean  $\frac{1}{g\rho}$  and therefore the segment around the mutator locus that has been free from recombination since the mutator variant arose has an expected length of  $\frac{2}{g\rho}$ . The expected number of additional mutations in a segment of this length is  $\frac{2\mu(\phi-1)}{\rho}$ . Note that this is independent of *g*, the age of the mutator allele. Taking  $\mu = 1.2 \times 10^{-8}$  [8],  $\rho = 1.26 \times 10^{-8}$  [21], gives an additional  $1.9(\phi - 1)$  mutations expected in a segment of mean length.

Whether it is feasible to detect a local peak in the density of derived alleles clearly depends on  $\phi$ , the affect of the mutator allele on the germline mutation rate. It also depends on the background number of derived alleles in haplotypes unaffected by the mutator allele. This, in turn, depends on *g* through the mean segment length. For example, if  $\phi$  is 5 and *g* is 20,000 generations, then the expected number of additional mutations in a segment of expected length is 7.6 and the mean segment length is approximately 8 Kb (this is likely an underestimate since it assumes uniform recombination). Assuming a pairwise heterozygosity rate in humans of 0.001 per bp [22] implies that we expect approximately 8 derived alleles per background haplotype in an 8 Kb segment. Although we expect a relatively small number of additional mutations in an affected haplotype, the signal of an excess in the number of derived alleles may be carried by several related haplotypes, such that the total number of unique excess mutations among the affected haplotypes may be much greater, potentially resulting in greater power to detect the affected loci.

#### Persistence of a mutator allele

Because the number of additional derived alleles is independent of the time since the mutator allele arose, but the segment length is not, the power to detect a mutator allele above the back-ground is greater for alleles that arose a large number of generations before the present. Given that the mutator allele is also required to be associated with a large effect on the germline mutation rate, it is necessary to consider whether such an allele could persist for a large number of generations despite the reduction in fitness that will result from an increased germline mutation rate. Although the increased burden of *de novo* mutations associated with the mutator allele is likely to reduce the fitness of individuals carrying it, this reduction may be relatively small [23], and the per generation germline mutation rate is capable of sustaining substantial variation in animals, for example ranging widely across mammalian species [2]. Bear in mind

also that *de novo* mutations will remain linked with the mutator allele only for an average of two generations before recombining away [2]. Indeed, a mutator allele, even at relatively low frequency, may contribute a large number of novel mutations, exerting a disproportionate influence on the evolution of the population as a whole while it persists.

The selective disadvantage associated with a (heterozygous) allele that increases the germline mutation rate can be approximated as  $2s_d \Delta U$  [2], where  $s_d$  is the mean selective disadvantage of a heterozygous deleterious mutation, U is the genomic deleterious mutation rate and  $\Delta U$  is the change in U resulting from the mutator allele. Thus given a value for U, a mutator allele that increases the germline mutation rate by a factor  $\phi$  will be associated with a selective coefficient (for heterozygotes) of  $s = -2s_d(\phi - 1)U$ .

The parameters  $s_d$  and U are not straightforward to estimate. Lynch [2] provides values for multicellular eukaryotes from the literature that range over orders of magnitude from  $10^{-3}$ – $10^{-2}$  for  $s_d$  and from 0.01–1 for U. For values at the lower ends of these ranges, even a mutator allele associated with a large value of  $\phi$  would have a relatively small effect on fitness. For example,  $\phi = 5$  gives s = -0.00008, which is in the region in which genetic drift begins to dominate over selection for an effective population size on the order of  $10^4$ . In this case a strong mutator allele could persist in a population for a large number of generations, sufficient to have an impact on the number of mutations that accumulate in the region linked to the mutator allele. By contrast, values of  $s_d$  and U at the higher ends of these ranges would give s = -0.08, indicating strong selection against a mutator allele. Such an allele would not persist for long and would rapidly reduce in frequency in an organism with even a moderate effective population size.

Estimates of *U* in particular genomic regions have been made by comparing between-species sequence divergence within these regions with divergence in putatively neutral regions (such as fourfold degenerate coding sites, pseudogenes or inactive transposable elements). Using this approach, Keightley [24] estimated a relatively high value of 2.2 deleterious mutations per generation across the whole genome in humans. There are caveats associated with this estimate (such as the assumption that genomic mutation rates are the same across sequence categories and possible bias from alignment in calculating genome-wide divergence), but it does seem likely that *U* is not substantially less than one.

The mean selective disadvantage is more difficult to estimate, due in part to the confounding influence of demographic factors. Boyko et al. [25] estimated values of around -0.03 for the mean selective disadvantage of heterozygous amino acid changing mutations in African Americans. However, this is likely to be unrepresentative of mutations genome wide and heavily influenced by strongly deleterious mutations. Also, although Boyko et al. [25] assumed a codominant selection model, many of these mutations are likely to be recessive and thus may not be deleterious when they arise de novo (as they are almost certain to be heterozygous). For Drosophila García-Dorado and Caballero [26] estimated a mean of 0.1 for the coefficient of dominance, h, defined such that the fitnesses of heterozygotes and homozygous mutants are 1 + hs and 1 + s respectively, and h = 0.5 corresponds to codominance. The best-fitting model of Boyko et al. [25] for the distribution of selective effects of amino acid changing mutations included a normally distributed component of weakly deleterious mutations with mean of -0.0002, as well as a point mass representing strongly deleterious mutations. Consistent with this, Do et al. [27] have shown that the absence of differential removal of deleterious mutations from the genomes of Africans and non-Africans implies that selection coefficients acting on non synonymous substitutions are either strong (s < -0.004) or very weak (s > -0.0004). In our context it may be possible to neglect strongly deleterious mutations on the grounds that they are typically recessive and rare outside protein-coding regions, where most de novo mutations occur. Mean selection coefficients for mutations outside protein coding regions are likely to be much smaller. If we set  $s_d = -0.0001$ , U = 1,  $\phi = 5$  and set h = 0.1 (to allow for the fact that even weakly deleterious mutations may have a coefficient of dominance well below 0.5), the selective disadvantage of the mutator allele in heterozygotes is approximately -0.0001. Such a mutation is affected by selection, but in a species with an effective population size similar to that of humans it can persist for a large number of generations by chance, particularly if it attains a high initial frequency due to a population bottleneck. The expected number of generations for which a mutation persists, under codominant selection, can be calculated [28]. For example a mutation with initial frequency of 0.2 and a selective coefficient of -0.0002 (-0.0001 in heterozygotes) has an expected time to extinction of 13,000 generations.

#### Proof of concept simulations

We used simulation to investigate the effect of a mutator allele on the numbers of derived alleles in the region in linkage disequilibrium with it. We considered a genomic region of 100 Kb, with a mutator locus at 50 Kb, and implemented simulations in a coalescent framework, with recombination and selection (see Materials and Methods for details). As expected, where mutations with a large impact on the germline mutation rate remained polymorphic in the population for a sufficient time, the genomic region close to the mutator locus was frequently found to exhibit a peak in the number of derived alleles, with this peak shared by a subset of haplotypes and the remaining haplotypes having a typical number of derived alleles across the region. An example of a simulation that exhibits such a peak is shown in Fig 1, and the results of ten simulation runs are provided as supplementary information S1 Fig. Whether a simulation displays this characteristic signature of a mutator locus depends on the history of recombination and coalescence in the surrounding genomic region. For example, recombination events close to the locus can disrupt linkage, eroding the increase in the number of derived alleles associated with the mutator locus. For the simulation shown in Fig 1 the mutator allele arose 20,000 generations before the present and resulted in a five-fold increase in the germline mutation rate in heterozygotes (and a ten-fold increase in homozygotes).

#### Genome scan for candidate mutator loci in humans

To search for human loci that may have carried mutator alleles, we obtained data from phase 3 of the G1K project [20]. The data consisted of inferred haplotypes for a total of 2504 individuals in 26 populations, along with putative ancestral alleles for each variable site inferred from Ensembl Compara release 59 [29]. For each haplotype in a population we counted the number of derived alleles in a sliding window of size 10 Kb along the genome. In each window we then calculated the maximum number of derived alleles in the window and the interquartile mean number of derived alleles across all haplotypes in the population (i.e. excluding the upper and lower 25th percentiles).

As expected, there was a strong linear correlation between the maximum and the interquartile mean (Fig 2). Our further expectation, supported by the above simulations, is that loci at which a mutator allele has existed in a population for a large number of generations can be identified from a characteristic pattern of variation, comprising a number of haplotypes with an unusually large number of derived alleles against a background of haplotypes with a typical number of derived alleles. Diversifying selection is likely to increase both the interquartile mean and the maximum number of derived alleles. By contrast, a low-frequency mutator allele should affect only the maximum, and consequently we excluded from consideration windows with large values of the interquartile mean (>75th percentile). The remaining windows were arranged in order of a statistic M, defined as the residual distance to the regression line (the latter shown in red in Fig 2), with windows showing the greatest



**Fig 1. Simulation study.** An example of simulated data showing a subset of haplotypes with a peak in the number of derived alleles in a 10 Kb sliding window. A region of 100 Kb was simulated over 40,000 generations using a coalescent approach with recombination. In this example a mutator allele with  $\phi = 5$  was introduced 20,000 generations before the present and was assumed to be weakly deleterious, with a selective coefficient of -0.0002. The red and green lines show the maximum and trimmed mean number of derived alleles in the window. Individual sampled haplotypes are shown in grey.

doi:10.1371/journal.pgen.1006549.g001

**PLOS** GENETICS

M-value considered as the best candidates for sites of ancient germline mutator alleles. We also removed candidate loci in which a high proportion of SNPs (>5%) were not in Hardy-Weinberg equilibrium, as potentially indicative of artefacts. The top 20 genomic regions by M-value are shown in Table 1.

To reduce the likelihood that divergent haplotypes that give rise to the entries in the table are the result mapping difficulties resulting from structural variants, we used the indel calls provided by the G1K project [30]. Two of the candidate loci in Table 1 (on chromosome 14 and at 192.6 Mb on chromosome 1) overlapped with a large structural variant (>50bp). The structural variant on chromosome 14 is very rare, occurring in just 15 haplotypes from the G1K Phase 3 data, while the highly derived haplotype at this locus is much more frequent



**Fig 2. Relationship between the maximum and the interquartile mean number of derived alleles.** A hexagon bin plot illustrating the relationship between the maximum and the interquartile mean number of derived alleles across all haplotypes in 10Kb windows from phase 3 of the G1K project. The data shown are for all populations combined, but the relationship is similar when populations are analyzed individually. The shading indicates the number of points within each hexagonal bin. The linear regression line is shown as a solid red line and the dashed red line identifies the candidate loci shown in Table 1. Note that each row of Table 1 can correspond to multiple points in the figure, due to multiple sliding windows overlapping the locus.

doi:10.1371/journal.pgen.1006549.g002

(occurring in 302 haplotypes). The opposite is the case for the structural variant at 192.6 Mb on chromosome 1, which is much more frequent (1871 haplotypes) than the highly derived haplotype at this locus (88 haplotypes). Therefore, in neither case is it plausible that mapping errors created by these structural variants are the cause of the observed signal. However, we

	GENETICS
--	----------

Chr	Position (Mb)	IQM	Max	Residual	Genes within 100 Kb	
1	202.4	15.4	106	84.28	UBE2T PPP1R12B	
18	15.3	12.2	42	24.1		
12	96.5	8.8	37	23.3	LTA4H ELK3	
2	18.0	15.4	45	23.3	MSGN1 GEN1 KCNS3 SMC6	
6	114.2	11.2	39	22.4	HDAC2 MARCKS	
11	1.4	12.1	40	22.2	BRSK2 TOLLIP MOB2 TOLLIP-AS1	
8	140.1	14.8	43	22.0		
16	32.4	13.5	41	21.5		
18	24.5	14.2	41	20.7	CHST9 AQP4	
12	87.4	12.2	38	20.1		
1	192.6	15.7	42	19.9	RGS1 MIR4426 RGS13	
12	99.9	13.5	39	19.6		
17	15.9	7.8	32	19.5	ADORA2B TTC19 NCOR1 ZSWIM7	
14	90.4	15.3	41	19.4	EFCAB11 TDP1 KCNK13	
6	130.9	11.9	36	18.6		
8	42.0	11.2	35	18.3	KAT6A PLAT AP3M	
10	25.0	14.6	39	18.2	ARHGAP21	
21	24.1	7.2	30	18.1		
12	44.6	7.2	30	18.1		
5	17.3	4.0	26	18.1	BASP1	

#### Table 1. Top 20 candidate mutator loci (relative to genome build hg19), ranked by residual.

#### IQM: interquartile mean

doi:10.1371/journal.pgen.1006549.t001

cannot rule out that unreported structural variants are the cause of some of the candidate mutator loci that we identified.

DNA repair genes are enriched at the candidate loci. Genes in proximity to the candidate loci shown in Table 1 are significantly enriched for just one cluster of overlapping biological processes, and encouragingly, these processes relate to DNA damage and repair. Five of these genes are annotated with the UniProt keywords DNA damage and DNA repair (p = 0.0004 and p = 0.0009, respectively), namely GEN1, SMC6, TDP1, UBE2T and ZSWIM7. Among these are genes that are annotated with the GO terms double-strand break repair via homologous recombination (GO:0000724; p = 0.005) and DNA repair (GO:0006281; p = 0.047). However, because genes are known to cluster on the genome by function, the above test of enrichment may be biased (for example, one candidate locus is responsible for two of the DNA-repair associated genes). To perform an unbiased test of enrichment of the candidate loci with DNA repair genes, we selected 20 loci at random and counted the number of DNA repair related genes within 100 Kb of these loci. Repeating this process 100,000 times, this number matched or exceeded the observed number 625 times (p = 0.006). This suggests that the enrichment of DNA repair genes is unlikely to be explained by functional clustering, and supports the existence of a subset of true positives among the list of candidate mutator alleles identified.

In addition to the known DNA repair genes, several other genes are plausible candidates for affecting the rate of germline mutation. *HDAC2* is a Class I histone deacetylase that has been shown to be enriched at replication forks [31] and to have a role in the DNA damage response, at least in the context of double strand break repair [32]. Mutations in *UBE2T* have recently been reported to cause Fanconi anemia [33], a condition characterised by genome

instability and cancer susceptibility that frequently results from defective DNA repair genes. *BRSK2* is included among 296 genes in the PathCards DNA repair superpath [34] and recently *MOB2* has also been strongly implicated in the DNA damage response [35].

**Demographic simulations.** We used demographic simulations to investigate the distribution of *M*-values in the absence of mutation rate polymorphism under different demographic scenarios (see Materials and Methods for details). As with the real data, the simulated data showed a strong linear relationship between the maximum (over haplotypes) derived allele count and the interquartile mean (S2 Fig). Two very different demographic scenarios were considered. In the first, a sample of size 2,500 (corresponding to the number of individuals in phase three of G1K) was taken from a single panmictic population. In the second, to explore the effects of population structure, the same number of individuals were sampled from a combined population resulting from pooling five separate populations that diverged 100 kyr ago. We found that both the relationship between interquartile mean and maximum number of derived alleles and the distribution of *M*-values were relatively insensitive to demography (S2 Fig). The simulations did not include selection and therefore generally had higher derived allele counts than the real data. Nonetheless, the most extreme value in the real data far exceeded the values found in the simulations (the most extreme *M*-values in the single population and multiple population simulation were 23.2 and 25.1, respectively). Although values as extreme as the remaining candidates in Table 1 were found in the simulated data, far more sliding windows exceeding the threshold for inclusion in Table 1 (residual  $\geq$ 18) were found in the real data than in the simulated data (there were 95 such 10 Kb sliding windows in the real data, compared to 11 and 16 in the single and five population simulations, respectively). We also investigated the effect of mutation clustering caused by multinucleotide mutation events by adding these to the coalescent simulation (see Materials and Methods). This had limited impact on the results (the number of windows exceeding the threshold rose from 11 to 13 in the single population simulations).

Thus we conclude that the signals observed at loci listed in Table 1 are unlikely to have arisen through the demographic and neutral evolutionary processes represented in our simulations, and more likely to result from alternative processes such as inter-individual variation in mutation rate.

#### Alternative explanations for the observed peaks

It is also necessary to consider other mechanisms, not included in our demographic simulations, that might have lead to excess divergence in a subset of haplotypes.

One example is a polymorphic gene conversion event affecting a subset of haplotypes, which could result in a block of divergent sequence. This would require gene conversion involving a segmentally duplicated region, in which the duplication of the target locus occurred prior to the divergence of humans from the other primates used to infer the ancestral state of genomic variants in this data. Under this scenario we may expect to find a subset of haplotypes that appear to be highly derived. A single gene conversion event giving rise to the diverged haplotypes would result in a phylogenetic tree of haplotypes consisting of one long internal branch separating two clades consisting of only short branches. The peak on chromosome 18 in Table 1 fits this description, but the remaining peaks do not (S3 Fig). For all of the remaining peaks, there is substantial diversity among the highly derived haplotypes, consistent with a greater rate of divergence than for the less highly derived haplotypes. Multiple gene conversion events would result in polyphyly of the diverged haplotypes, and this is not observed for most of the peaks; however, accurate phylogenetic inference may be exceptionally difficult in the gene conversion scenario.

Another possible mechanism might be an extreme multi-nucleotide mutation event, more substantial than those included in our simulations, giving rise to multiple mutations simultaneously. However, similarly to gene conversion, this would also create two clades, both consisting of short branches, separated by a long branch that reflects the multi nucleotide mutation. Thus for trees of this nature we cannot rule out either gene conversion or extreme multi-nucleotide mutation. However we note that such trees are also consistent with the effects of a mutator allele, since a relatively recent coalescence of the highly derived haplotypes would also produce a long internal branch separating clades consisting mostly of short branches. Indeed, this was frequently observed in the proof-of-concept simulations (S4 Fig).

Introgression of haplotypes derived from hybridisation between modern humans and Neanderthals as well as other ancient hominins has been reported [36–39]. However, this introgression is not likely to result in a subset of haplotypes with an excess of derived alleles. Alleles were designated as derived or ancestral by comparison to other primates and the number of derived alleles on a given Neanderthal haplotype should be similar to the number for a haplotype from modern humans. Moreover, almost all of the peaks in Table 1 are found in Africans (S5 Fig), but only non-Africans have Neanderthal ancestry. The one exception is again the peak on chromosome 18. Indeed, for this peak we found that the highly derived haplotypes are enriched for haplotypes of likely Neanderthal origin (as determined by [40]). This peak is not close to any DNA repair genes and its removal from the list of candidates would in fact slightly increase the statistical significance of the association with DNA repair genes discussed above.

As a further potential alternative explanation, we also considered the case of a genomic segmental duplication segregating within the population, which could give rise to mapping errors in individuals carrying it, such that derived alleles called in the candidate locus are due to incorrectly mapped sequence reads. However, a simple duplication could cause at most an apparent doubling of divergence on affected haplotypes in this way, and therefore would be unlikely to be included as a candidate under the thresholds we have applied. A more complicated tandem duplication with several copies might give rise to higher apparent divergence, but such sites are very unlikely to have passed the filtering criteria used in the G1K calling protocol.

The peaks we observe in the derived allele count could potentially be caused by extremely strong selective pressure acting on a subset of populations from the G1K dataset. Such strong and localised selective pressure could cause a population or group of populations to have a large number of derived alleles within a genomic region, while the remainder of the populations in the dataset that did not experience the strong selective pressure have a typical number of derived alleles. Even if selective pressures strong enough to bring multiple linked derived alleles to high frequencies had acted on human populations we find no evidence for this hypothesis in the data, as, for most of the peaks, the highly derived haplotypes are found in human populations from multiple populations (S5 Fig).

Our proof-of-concept simulations suggest that most of the peaks in Table 1 are consistent with the effects of mutator alleles with large effect size which were maintained in the population for a large number of generations. The difference between the maximum and inter-quartile mean number of derived alleles in Table 1 is below 30 for all but the first locus, and in the low twenties for loci towards the bottom of the table. The proof-of-concept simulations also had large differences in the maximum and interquartile mean when the signal was detectable (just over 20 for simulations 8 and 9 in S1 Fig). The top locus in the table, however, has a much larger difference between the maximum and interquartile mean number of derived alleles. Although we could not find an alternative explanation for this signal, it seems difficult to reconcile with the effects of a mutator allele whose effect size would have enabled it to persist in the population for a large number of generations.

#### Support from human trio data

Although we have considered the approach described here primarily as a means of detecting the remnant signatures of ancient mutator alleles, if such an allele were still active and linked to a highly-derived haplotype, then it might be possible to find supporting evidence in whole genome sequence data from trios. In principle, an active a mutator allele would give rise to an association between the presence of the highly-derived haplotype in the parents and an elevated number of *de novo* mutations in the offspring.

We investigated this possibility for the candidate loci shown in Table 1 by obtaining counts of *de novo* mutations in the offspring of complete trios and genome-wide parental genotype data from the G1K (n = 59) and GoNL (n = 248) projects [19, 20]. For each of the loci shown in Table 1 we tested for a positive association between the number of *de novo* mutations in the offspring and the presence in the corresponding parent of the highly derived haplotypes at the putative mutator locus (Table 2). P-values from two tests are reported for each peak, one based on fitting a robust linear model and the second based on permutation (referred to as  $p_t$  and  $p_{perm}$  in Table 2, respectively; see Materials and Methods for details). The analysis was carried out separately for male and female parents and for the two projects, to avoid biases resulting from differences in study methodology confounded with differences in populations of origin. In each case we required at least five parents with a highly derived haplotype to perform the test.

In the case of the G1K dataset 21 tests were performed (10 paternal and 11 maternal). Of these, two showed a nominally significant positive association between the number of *de novo* mutations and the presence of the highly derived haplotype in the parent in both statistical tests (Table 2). The most significant of these was for the peak just upstream of *BRSK2* on chromosome 11 (illustrated in Fig 3). With the GoNL dataset, 15 tests could be performed (seven paternal and eight maternal), but no significant associations were observed (Table 2). Considering that 36 tests in total were carried out in total, the associations in Table 2 do not remain significant following Bonferroni correction, and additional diverse trio datasets may be necessary to confirm these associations definitively. Data forming the basis for the tests shown in Table 2 are provided as S1 and S2 Tables.

Paternal age at conception is strongly correlated with the number of *de novo* mutations in offspring [8]; however, paternal age was not available for the G1K data, while for the G0NL data we included the ages of both parents at conception in the linear model relating *de novo* mutation count to the presence of the highly derived haplotype. There remained no significant positive associations in the case of this dataset. For the G1K dataset population of origin information was available for the samples; however, there was no statistically significant relationship between population of origin and the *de novo* mutation count (S6 Fig).

Interestingly, the candidate locus that shows the strongest evidence of an association with the *de novo* mutation count in the G1K trios is just upstream of *BRSK2* (Fig 3b), a highly conserved serine threonine kinase which is preferentially expressed in brain and testis and shows enhanced activity in response to DNA damage [41, 42]. The observed association was *maternal* rather than paternal; however, expression in testis is consistent with a role in germline DNA replication.

At the same time, this haplotype is not associated with an elevated *de novo* mutation count in the GoNL data (Table 2). A possible explanation for this is that despite its much larger sample size, the GoNL cohort, which comprises individuals of Dutch ancestry, is substantially less diverse than the G1K cohort. As noted above, although the highly derived haplotype was observed in parents of trios in the GoNL cohort, its presence does not necessarily imply the presence of the causative mutator allele: a recombination event on the ancestral lineage leading

Chr	Position (Mb)	G1K Paternal	G1K Maternal	GoNL Paternal	GoNL Maternal
1	202.4	NA	NA	NA	NA
18	15.3	NA	NA	NA	NA
12	96.5	NA	NA	NA	NA
2	18	0.49, 0.48	0.86, 0.88	NA	0.26, 0.26
6	114.2	NA	NA	NA	NA
11	1.4	0.20, 0.21	0.0033, 0.0039	0.95, 0.95	0.91, 0.91
8	140.1	0.75, 0.75	0.29, 0.30	0.16, 0.16	0.13, 0.13
16	32.4	NA	0.034, 0.067	NA	NA
18	24.5	NA	NA	NA	NA
12	87.4	NA	NA	0.49, 0.48	0.41, 0.41
1	192.6	0.049, 0.064	0.24, 0.26	NA	NA
12	99.9	NA	NA	NA	NA
17	15.9	NA	NA	NA	NA
14	90.4	0.63, 0.62	0.55, 0.55	0.75, 0.76	0.96, 0.97
6	130.9	NA	NA	NA	NA
8	42.0	0.26, 0.26	0.59, 0.59	NA	NA
10	25.0	0.49, 0.21	0.56, 0.57	0.45, 0.44	0.13, 0.13
21	24.1	0.67, 0.68	0.51, 0.51	NA	NA
12	44.6	0.024, 0.040	0.44, 0.44	0.43, 0.42	0.10, 0.10
5	17.3	0.22, 0.23	0.70, 0.70	1.0, 1.0	0.87, 0.87

Table 2. P-values ( $p_{t}$ ,  $p_{perm}$ ) from tests of elevated rate of *de novo* mutation in trios.

doi:10.1371/journal.pgen.1006549.t002

to the highly derived haplotype in the Dutch population may have separated the two. Indeed, in this way it is possible for an extinct mutator allele to leave a fossil peak in the derived allele count of a contemporary population, and it is plausible that many of the signals we have detected are of this nature.



**Fig 3.** Putative mutator allele on chromosome 11. Relationship between the number of *de novo* mutations in the offspring and the maternal number of highly derived haplotypes of the putative mutator allele on chromosome 11 (a). Location of the putative mutator locus on chromosome 11, defined as a peak in the difference between the maximum and interquartile mean number of derived alleles across haplotypes (b).

doi:10.1371/journal.pgen.1006549.g003

The candidate mutator locus on chromosome 14 is also noteworthy. This peak overlaps precisely with an exon of the tyrosyl-DNA phosphodiesterase 1 (*TDP1*) gene (S7 Fig). The product of *TDP1* has a role in the repair of stalled topoisomerase I-DNA complexes. This candidate is not supported by the available trio datasets, despite the highly derived haplotypes being found in sufficient numbers in both cohorts, suggesting that its occurrence on the *TDP1* gene is either a striking coincidence or that it may be a fossilised peak resulting from a mutator allele that is extinct or at least much rarer than the highly derived haplotypes to which it has given rise.

#### Conclusions

We have shown that when a genomic locus contains a polymorphism in which one allele increases the germline mutation rate, a characteristic signature may result, comprising a subset of haplotypes that are locally more divergent than other haplotypes at that locus. Whether this signature can be detected depends on the magnitude of the effect and on the time since the mutator allele arose. The number of additional mutations on an unrecombined segment of expected length carrying the mutator allele is independent of the time since the mutator allele arose, but the likelihood of detecting a fixed number of additional mutations over the background increases as the segment length decreases and is thus higher for older mutations.

We searched for such signatures in data from Phase 3 of the 1000 Genomes Project, and found candidate human genetic loci that may have once contained (and in some cases may still contain) germline mutator alleles. Consistent with this expectation, the set of genes located in the vicinity of these loci is enriched for genes involved in DNA repair. A test for the continued presence of active mutator alleles in these haplotypes, by looking for association with high numbers of *de novo* mutations passed to offspring in two trio sequencing cohorts, found support for association in two cases, but no associations with overall significance.

It may therefore be that mutator alleles are no longer present, or are now very rare, in most of the candidate loci we identified. However the trio cohorts we examined were small relative to what may be feasible in future studies, and in one case (the GoNL data) represented a population which is comparatively homogeneous in terms of its genetic ancestry. Further trio cohorts from more genetically diverse populations may be necessary to test some of the candidates we identified. Such studies will also shed light on the full extent of polymorphism in the rate of *de novo* mutation in human populations.

It is difficult to conclusively exclude alternative explanations for the signals we have detected. However we have considered several such explanations and have shown that they are either far less likely to generate signals of this nature (in the case of neutral demographic processes) or would generate loci with characteristics not seen in these cases (for example in the case of a mis-mapped segmental duplication).

More generally, it is far from implausible *a priori* that there are genetic factors affecting the germline mutation rate in humans, and that there may arise from time to time mutator alleles with a moderately large effect size. We have also shown that such alleles can plausibly persist for many generations and leave a detectable signal that survives after they have disappeared. It may also be that different such alleles affect the mutation process in different ways, and that as well as changing the overall germline mutation rate they give rise to different mutational signatures. Indeed recent studies have found variation in the mutational spectrum (expressed in terms of the relative frequency of different triplet sequence contexts for de novo mutations) between human populations and over time, including evidence for past activity of mutator alleles that are now rare or extinct ([15, 43]).

On longer timescales, comparison of mutation rates estimated from inter-specific divergence (e.g. [44]) with measurements of the mutation rate in contemporary populations points to a slowdown in human and great ape evolution [45]. Some of this may be due to changes in life history parameters such as generation time; however, there may also have been changes in the underlying per-generation mutation rate, or more specifically, changes in cellular mutation rates during development and gametogenesis, perhaps driven by genetic variation at loci such as those detected here. Indeed, it is worth considering that the higher mutation rate inferred for some haplotypes at candidate loci may represent the ancestral rather than derived state, such that the mutations in question resulted in anti-mutator rather than mutator alleles.

#### **Materials and Methods**

#### Proof of concept simulations

We used a coalescent approach to simulate the effects of a mutation that increases the germline mutation rate. The mutator allele is introduced at an initial frequency  $p_0$  at time  $t_0$  and we assume no mutation between the mutator allele and wild-type after time  $t_0$ . We use a constant effective population size of 10,000 individuals and incorporate recombination, occurring at a constant rate across the simulated genomic region (see below for simulations of null models with variable recombination rate and demographics). The mutator allele is associated with a selective coefficient, s, and we incorporate selection against the mutator allele using an approach described by Kaplan et al. [46]. We first obtained the deterministic allele frequency trajectory in the forward direction, with s = -0.0002. Because we require that chromosomes carrying the mutator allele have an elevated mutation rate the simulations could not be performed using existing implementations of the coalescent with recombination and selection (e.g. [47]) and were instead implemented as a Perl script, which is available from the authors, on request. To allow for the effects of the mutator allele on the rate of mutation of the homologous chromosome in heterozygous individuals, at each generation the mutation rate was elevated for chromosomes that did not carry the mutator allele with a probability equal to the frequency of the mutator allele in that generation.

#### Demographic simulations

To examine the genome-wide distribution of maximum and inter-quartile mean number of derived alleles per window, we simulated populations evolving under a model of neutral evolution and without mutation rate polymorphism. Two simulations of 5000 haploid whole genome sequences were carried out, one comprising a single population with constant effective population size, and another comprising five separate populations diverging 100 kyr ago, to explore the potential effects of population structure on the signal of interest. All populations simulated were of equal size with a scaled mutation rate  $\theta = 0.001$ , matching the mean genome-wide heterozygosity in modern human populations. Simulations were carried out using the coalescent simulator msprime [48], with a variable recombination rate matching that found in the human genome (including recombination hotspots) [49], and a uniform genome-wide mutation rate.

#### Genome scan

Simulations suggested that a mutation that increases the rate of germline mutation and is maintained in a population for a sufficiently long time can result in a genomic region in which haplotypes linked to the mutator allele have a large number of derived alleles and the remaining haplotypes have typical numbers of derived alleles (i.e. the effect of the mutator

allele on unlinked haplotypes is the same as its effect on the rest of the genome). Variant calls in .vcf format were downloaded for phase 3 (release v5a) of the 1,000 Genomes project [20]. Putative ancestral alleles (derived by the 1,000 Genomes project from Ensembl Compara release 59 [29]) were obtained from the vcf files, restricting to SNPs with high confidence ancestral alleles. Using a sliding window (window size: 10 Kb; step size: 1 Kb) we calculated the interquartile mean (also called the 25% trimmed mean) and the maximum number of derived alleles in each window. There was a linear relationship between the maximum and the interquartile mean, reflecting the fact that windows within which the interquartile mean number of derived alleles was large also had a large value for the maximum number of derived alleles (over all haplotypes). We fitted linear regression models to the maximum, as a function of the interquartile mean and calculated the distance from the regression line for each window. Windows with an elevated value of the interquartile mean (>75th percentile) were excluded and the remaining windows were sorted by distance to the regression line.  $\chi^2$ tests of Hardy-Weinberg equilibrium (at alpha = 0.05) were carried out separately within each population for each SNP within the candidate regions. Windows within which a high proportion (>5%) of SNPs rejected Hardy-Weinberg equilibrium were removed. This resulted in the removal of nine loci.

#### Trio validation

We obtained *de novo* mutation counts in offspring and parent variant call data for human parent-offspring trios from the G1K (n = 59) and GoNL (n = 248) projects [19, 20] (see [43]) for details of methods used to call de novo mutations from the G1K trios). For each candidate mutator locus we determined the number (0, 1 or 2) of copies of the highly derived haplotype in each parent. For the phased G1k data we plotted the number of derived alleles per haplotype and if this distribution included a subset of highly derived haplotypes we selected a threshold that gave the best separation between the highly derived and typically derived haplotypes (S8 Fig) and used this to classify each haplotype. For unphased data (GoNL) we plotted the number of derived alleles per individual. When the highly derived haplotypes occurred at low frequency this distribution was typically bimodal (all individuals had 0 or 1 highly derived haplotype), becoming trimodal when the highly derived haplotypes occurred at higher frequency (corresponding to individuals with 0, 1 or 2 highly derived haplotypes). In each case we set a threshold number of derived alleles that gave the best separation of the peaks in the distribution of the derived allele count (S9 Fig). We then tested for an association between the offspring *de novo* mutation count and the number of copies of the highly derived haplotype by fitting robust linear regression models using the rlm function from the MASS package [50] in R [51], with the default M-estimation method. The G1K and GoNL trios were analysed separately to avoid the potential for confounding between study methodologies and population of origin and results are reported separately for each parent (to enable the detection of parent-of-origin specific effects). Parental age at conception was available for the GoNL trios and included as a covariate in the linear models. P-values for the null hypothesis that the regression coefficient for the offspring de novo mutation count as a function of the number of copies of the highly derived haplotype was not greater than zero were calculated in two ways: from the right tail of the distribution of the regression coefficient tstatistic (referred to as  $p_t$  in Table 2) and by permutation. For the latter we performed 10,000 permutations of the *de novo* mutation counts and determined the proportion of times the tstatistic from the robust regression for the permuted data was greater than or equal to the value for the unpermuted data (reported as  $p_{perm}$  in Table 2). Correction for multiple testing was carried out using the Bonferroni method.

#### Enrichment of DNA repair genes close to candidate loci

We identified all genes within 100 Kb of each of the candidate mutator loci (excluding pseudogenes) and tested for functional enrichment using DAVID [52, 53], version 6.8 with default parameters (on 3/11/2016). Given that there were several genes found within 100 Kb of many of the candidate mutator loci the phenomenon of functional gene clustering means that this set of genes is not independent, creating a potential for bias in the statistical tests of enrichment. We therefore also devised a test based on randomisation of the genomic loci. For this test 20 loci (equal to the number of loci in Table 1) were sampled at random from the regions of the data for which we had variant call data and the number of DNA repair genes (annotated with GO term GO:0006281 in Ensembl 84) within 100 Kb of these random loci was determined. This was repeated 100,000 times and the numbers of DNA repair genes in the random data was compared to the observed number of genes with this GO term within 100 Kb of the candidate mutator loci.

### **Supporting Information**

**S1 Fig. Proof of concept simulations.** Ten random simulations, equivalent to the simulation shown in Fig 1.

(PDF)

**S2 Fig. Demographic simulations.** (a,b) Hexagon bin plots illustrating the relationship between the maximum and the interquartile mean number of derived alleles across all haplo-types in 10Kb windows for two simulated demographic scenarios, consisting of a single population (a) or five populations separated 100 kya (b). The shading indicates the number of points within each hexagonal bin. The linear regression line is shown as a solid red line and the dashed red line corresponds to the threshold for the candidate loci shown in Table 1. (c) Quantile-quantile plot of the residuals from the regression line in the single population and five-populations. (PDF)

**S3 Fig. Plots of the 20 candidate mutator loci shown in Table 1.** For each candidate mutator locus in Table 1 we show a phylogenetic tree (left), constructed from a 10 Kb window overlapping the peak. Due to the difficulty displaying all 5008 haplotypes from the 1000 Genomes Project we sampled 100 haplotypes at random and supplemented these with up to approximately 50 of the highly derived haplotypes, again at random. Bottom right panels illustrate for this reduced set of haplotypes the number of derived (red) and ancestral alleles (blue) in each haplotype across the 10Kb window. Each haplotype is represented as a row on the plot and the haplotypes with the largest number of derived alleles are grouped at the top of the plot. Tick marks indicate the positions of individual SNPs. The top right panels show the maximum (red) and interquartile mean (green) number of derived alleles, as well as the number of derived alleles on actual haplotypes (grey). (PDF)

**S4 Fig. Trees from proof-of-concept simulations.** Trees from the subset of the ten proof-of-concept simulations that showed a peak in the maximal derived allele count. (PDF)

**S5 Fig. Population of origin of the highly derived haplotypes.** The proportion of the highly derived haplotypes at each peak corresponding to populations from Africa, Asia, America and Europe. The rightmost bar shows the proportions of each continent of origin among all G1K phase 3 samples. (PDF)

**S6 Fig. Population of origin effect on** *de novo* **mutation count.** Stripchart showing *de novo* mutation count as a function of population of origin in the G1K data. (PDF)

**S7 Fig. Candidate mutator locus within the** *TDP1* gene. (PDF)

S8 Fig. Derived allele count thresholds used to decide highly derived haplotypes for the G1K data.

(PDF)

**S9** Fig. Derived allele count thresholds used to decide the number of highly derived haplotypes in each individual for the unphased GoNL data. (PDF)

**S1 Table. Trio data from 1000 Genomes.** The number of de novo mutations and number of copies of the highly derived haplotype in trios from the 1000 Genomes Project. (TXT)

**S2 Table. Trio data from the Genome of the Netherlands Project.** The number of de novo mutations and estimated number of copies of the highly derived haplotype in trios from the Genome of the Netherlands Project.

(TXT)

#### Acknowledgments

This study makes use of data generated by the Genome of the Netherlands Project and the 1000 Genomes Project. We thank the scientists and participants of the 1000 Genomes and Genome of the Netherlands projects for generously sharing their data.

#### **Author Contributions**

Conceptualization: CS.

Formal analysis: CS AS.

Investigation: CS AS.

Methodology: CS.

Visualization: CS.

Writing – original draft: CS.

Writing - review & editing: CS AS.

#### References

- 1. Baer CF, Miyamoto MM, Denver DR. Mutation rate variation in multicellular eukaryotes: causes and consequences. Nat Rev Genet. 2007; 8(8):619–631. doi: 10.1038/nrg2158 PMID: 17637734
- 2. Lynch M. Evolution of the mutation rate. Trends Genet. 2010; 26(8):345–352. doi: 10.1016/j.tig.2010. 05.003 PMID: 20594608
- Haldane JBS. The rate of spontaneous mutation of a human gene. J Genet. 1935; 31:317–326. doi: 10. 1007/BF02982403
- 4. Kondrashov AS. Direct estimates of human per nucleotide mutation rates at 20 loci causing Mendelian diseases. Hum Mutat. 2003; 21(1):12–27. doi: 10.1002/humu.10147 PMID: 12497628

- Lynch M. Rate, molecular spectrum, and consequences of human mutation. Proc Natl Acad Sci USA. 2010; 107(3):961–968. doi: 10.1073/pnas.0912629107 PMID: 20080596
- Roach JC, Glusman G, Smit AF, Huff CD, Hubley R, Shannon PT, et al. Analysis of genetic inheritance in a family quartet by whole-genome sequencing. Science. 2010; 328(5978):636–639. doi: 10.1126/ science.1186802 PMID: 20220176
- Conrad DF, Keebler JE, DePristo MA, Lindsay SJ, Zhang Y, Casals F, et al. Variation in genome-wide mutation rates within and between human families. Nat Genet. 2011; 43(7):712–714. doi: 10.1038/ng. 862 PMID: 21666693
- Kong A, Frigge ML, Masson G, Besenbacher S, Sulem P, Magnusson G, et al. Rate of de novo mutations and the importance of father's age to disease risk. Nature. 2012; 488(7412):471–475. doi: 10. 1038/nature11396 PMID: 22914163
- 9. Goldmann JM, Wong WS, Pinelli M, Farrah T, Bodian D, Stittrich AB, et al. Parent-of-origin-specific signatures of de novo mutations. Nat Genet. 2016; 48(8):935–939. doi: 10.1038/ng.3597 PMID: 27322544
- Segurel L, Wyman MJ, Przeworski M. Determinants of Mutation Rate Variation in the Human Germline. Annu Rev Genomics Hum Genet. 2014;. doi: 10.1146/annurev-genom-031714-125740 PMID: 25000986
- Jiang YH, Yuen RK, Jin X, Wang M, Chen N, Wu X, et al. Detection of clinically relevant genetic variants in autism spectrum disorder by whole-genome sequencing. Am J Hum Genet. 2013; 93(2):249–263. doi: 10.1016/j.ajhg.2013.06.012 PMID: 23849776
- Michaelson JJ, Shi Y, Gujral M, Zheng H, Malhotra D, Jin X, et al. Whole-genome sequencing in autism identifies hot spots for de novo germline mutation. Cell. 2012; 151(7):1431–1442. doi: <u>10.1016/j.cell.</u> 2012.11.019 PMID: 23260136
- Harris K. Evidence for recent, population-specific evolution of the human mutation rate. Proc Natl Acad Sci USA. 2015; 112(11):3439–3444. doi: 10.1073/pnas.1418652112 PMID: 25733855
- 14. Mathieson I, Reich DE. Variation in mutation rates among human populations. bioRxiv. 2016;http://dx. doi.org/10.1101/063578.
- Harris K, Pritchard J. Rapid evolution of the human mutation spectrum. bioRxiv. 2016;http://dx.doi.org/ 10.1101/084343.
- Aarnio M, Sankila R, Pukkala E, Salovaara R, Aaltonen LA, de la Chapelle A, et al. Cancer risk in mutation carriers of DNA-mismatch-repair genes. Int J Cancer. 1999; 81(2):214–218. doi: 10.1002/(SICI) 1097-0215(19990412)81:2%3C214::AID-IJC8%3E3.0.CO;2-L PMID: 10188721
- Briggs S, Tomlinson I. Germline and somatic polymerase 
  *ϵ* and *δ* mutations define a new class of hypermutated colorectal and endometrial cancers. J Pathol. 2013; 230(2):148–153. doi: 10.1002/path.4185 PMID: 23447401
- de la Chapelle A. Genetic predisposition to colorectal cancer. Nat Rev Cancer. 2004; 4(10):769–780. doi: 10.1038/nrc1453 PMID: 15510158
- Francioli LC, Polak PP, Koren A, Menelaou A, Chun S, Renkens I, et al. Genome-wide patterns and properties of de novo mutations in humans. Nat Genet. 2015; 47(7):822–826. doi: <u>10.1038/ng.3292</u> PMID: 25985141
- Auton A, Brooks LD, Durbin RM, Garrison EP, Kang HM, Korbel JO, et al. A global reference for human genetic variation. Nature. 2015; 526(7571):68–74. doi: 10.1038/nature15393 PMID: 26432245
- Jensen-Seaman MI, Furey TS, Payseur BA, Lu Y, Roskin KM, Chen CF, et al. Comparative recombination rates in the rat, mouse, and human genomes. Genome Res. 2004; 14(4):528–538. doi: 10.1101/gr. 1970304 PMID: 15059993
- Li H, Durbin R. Inference of human population history from individual whole-genome sequences. Nature. 2011; 475(7357):493–496. doi: 10.1038/nature10231 PMID: 21753753
- Lynch M. The lower bound to the evolution of mutation rates. Genome Biol Evol. 2011; 3:1107–1118. doi: 10.1093/gbe/evr066 PMID: 21821597
- 24. Keightley PD. Rates and fitness consequences of new mutations in humans. Genetics. 2012; 190 (2):295–304. doi: 10.1534/genetics.111.134668 PMID: 22345605
- Boyko AR, Williamson SH, Indap AR, Degenhardt JD, Hernandez RD, Lohmueller KE, et al. Assessing the evolutionary impact of amino acid mutations in the human genome. PLoS Genet. 2008; 4(5): e1000083. doi: 10.1371/journal.pgen.1000083 PMID: 18516229
- Garcia-Dorado A, Caballero A. On the average coefficient of dominance of deleterious spontaneous mutations. Genetics. 2000; 155(4):1991–2001. PMID: 10924491
- Do R, Balick D, Li H, Adzhubei I, Sunyaev S, Reich D. No evidence that natural selection has been less effective at removing deleterious mutations in Europeans than in West Africans. arXiv:14024896. 2014;.

- 28. Ewens WJ. Mathematical Population Genetics I. Theoretical Introduction. New York: Springer; 2004.
- Flicek P, Amode MR, Barrell D, Beal K, Billis K, Brent S, et al. Ensembl 2014. Nucleic Acids Res. 2014; 42(Database issue):D749–755. doi: 10.1093/nar/gkt1196 PMID: 24316576
- Sudmant PH, Rausch T, Gardner EJ, Handsaker RE, Abyzov A, Huddleston J, et al. An integrated map of structural variation in 2,504 human genomes. Nature. 2015; 526(7571):75–81. doi: 10.1038/ nature15394 PMID: 26432246
- **31.** Sirbu BM, Couch FB, Feigerle JT, Bhaskara S, Hiebert SW, Cortez D. Analysis of protein dynamics at active, stalled, and collapsed replication forks. Genes Dev. 2011; 25(12):1320–1327. doi: 10.1101/gad. 2053211 PMID: 21685366
- Miller KM, Tjeertes JV, Coates J, Legube G, Polo SE, Britton S, et al. Human HDAC1 and HDAC2 function in the DNA-damage response to promote DNA nonhomologous end-joining. Nat Struct Mol Biol. 2010; 17(9):1144–1151. doi: 10.1038/nsmb.1899 PMID: 20802485
- Hira A, Yoshida K, Sato K, Okuno Y, Shiraishi Y, Chiba K, et al. Mutations in the gene encoding the E2 conjugating enzyme UBE2T cause Fanconi anemia. Am J Hum Genet. 2015; 96(6):1001–1007. doi: 10. 1016/j.ajhg.2015.04.022 PMID: 26046368
- Belinky F, Nativ N, Stelzer G, Zimmerman S, Iny Stein T, Safran M, et al. PathCards: multi-source consolidation of human biological pathways. Database (Oxford). 2015; 2015. doi: <u>10.1093/database/</u> bav006
- Gomez V, Gundogdu R, Gomez M, Hoa L, Panchal N, O'Driscoll M, et al. Regulation of DNA damage responses and cell cycle progression by hMOB2. Cell Signal. 2015; 27(2):326–339. doi: <u>10.1016/j.</u> cellsig.2014.11.016 PMID: 25460043
- Green RE, Krause J, Briggs AW, Maricic T, Stenzel U, Kircher M, et al. A draft sequence of the Neandertal genome. Science. 2010; 328(5979):710–722. doi: 10.1126/science.1188021 PMID: 20448178
- Sankararaman S, Mallick S, Dannemann M, Prufer K, Kelso J, Paabo S, et al. The genomic landscape of Neanderthal ancestry in present-day humans. Nature. 2014; 507(7492):354–357. doi: <u>10.1038/</u> nature12961 PMID: 24476815
- Prufer K, Racimo F, Patterson N, Jay F, Sankararaman S, Sawyer S, et al. The complete genome sequence of a Neanderthal from the Altai Mountains. Nature. 2014; 505(7481):43–49. doi: 10.1038/ nature12886 PMID: 24352235
- Vernot B, Akey JM. Resurrecting surviving Neandertal lineages from modern human genomes. Science. 2014; 343(6174):1017–1021. doi: 10.1126/science.1245938 PMID: 24476670
- 40. Sankararaman S, Mallick S, Dannemann M, Prufer K, Kelso J, Paabo S, et al. The genomic landscape of Neanderthal ancestry in present-day humans. Nature. 2014; 507(7492):354–357. doi: 10.1038/ nature12961 PMID: 24476815
- Lu R, Niida H, Nakanishi M. Human SAD1 kinase is involved in UV-induced DNA damage checkpoint function. J Biol Chem. 2004; 279(30):31164–31170. doi: 10.1074/jbc.M404728200 PMID: 15150265
- Sabater L, Gomez-Choco M, Saiz A, Graus F. BR serine/threonine kinase 2: a new autoantigen in paraneoplastic limbic encephalitis. J Neuroimmunol. 2005; 170(1–2):186–190. doi: 10.1016/j.jneuroim. 2005.08.011 PMID: 16165222
- 43. Narasimhan VM, Rahbari R, Scally A, Wuster A, Mason D, Xue Y, et al. A direct multi-generational estimate of the human mutation rate from autozygous segments seen in thousands of parentally related individuals. bioRxiv. 2016;http://dx.doi.org/10.1101/059436.
- 44. Nachman MW, Crowell SL. Estimate of the mutation rate per nucleotide in humans. Genetics. 2000; 156(1):297–304. PMID: 10978293
- 45. Scally A. The mutation rate in human evolution and demographic inference. Curr Opin Genet Dev. 2016; 41:36–43. doi: 10.1016/j.gde.2016.07.008 PMID: 27589081
- Kaplan NL, Hudson RR, Langley CH. The "hitchhiking effect" revisited. Genetics. 1989; 123(4):887– 899. PMID: 2612899
- Ewing G, Hermisson J. MSMS: a coalescent simulation program including recombination, demographic structure and selection at a single locus. Bioinformatics. 2010; 26(16):2064–2065. doi: 10.1093/ bioinformatics/btq322 PMID: 20591904
- Kelleher J, Etheridge AM, McVean G. Efficient Coalescent Simulation and Genealogical Analysis for Large Sample Sizes. PLoS Comput Biol. 2016; 12(5):1–22. doi: 10.1371/journal.pcbi.1004842
- 49. Frazer KA, Ballinger DG, Cox DR, Hinds DA, Stuve LL, Gibbs RA, et al. A second generation human haplotype map of over 3.1 million SNPs. Nature. 2007; 449(7164):851–861. doi: <u>10.1038/nature06258</u> PMID: <u>17943122</u>
- Venables WN, Ripley BD. Modern Applied Statistics with S. 4th ed. New York: Springer-Verlag New York; 2002.

- 51. R Core Team. R: A Language and Environment for Statistical Computing; 2015. Available from: https:// www.R-project.org/.
- 52. Huang daW, Sherman BT, Lempicki RA. Systematic and integrative analysis of large gene lists using DAVID bioinformatics resources. Nat Protoc. 2009; 4(1):44–57. doi: 10.1038/nprot.2008.211
- 53. Huang daW, Sherman BT, Lempicki RA. Bioinformatics enrichment tools: paths toward the comprehensive functional analysis of large gene lists. Nucleic Acids Res. 2009; 37(1):1–13. doi: 10.1093/nar/gkn923